Author’s response to reviews

Title: Race and region have independent and synergistic effects on dietary intakes in black and white women

Authors:

P. K. Newby (pknewby@post.harvard.edu)
Sabrina E Noel (sabrina.feldeisen@gmail.com)
Rachael Grant (rkg43@cornell.edu)
Suzanne Judd (sjudd@ms.soph.uab.edu)
James M Shikany (jshikany@mail.dopm.uab.edu)
Jamy Ard (ardj@uab.edu)

Version: 4 Date: 5 March 2012

Author’s response to reviews:

Please see the following letter to the editors.